Author’s response to reviews

Title: Efficacy of Inhaled Budesonide on Prevention of Acute Mountain Sickness during Emergent Ascent: A Meta-analysis of Randomized Controlled Trials

Authors:

Gaurav Nepal (gauravnepal@iom.edu.np)
Jayant Yadav (jkyadav22@iom.edu.np)
Jessica Rehrig (jrehrig@une.edu)
Niroj Bhandari (bhandariniroj15@gmail.com)
Santosh Baniya (baniyasantosh@iom.edu.np)
Rakesh Ghimire (drrakesghimire@iom.edu.np)
Narayan Mahotra (narayanmahotra@gmail.com)

Version: 2 Date: 07 Apr 2020

Author’s response to reviews:

Dear Richard Neville Bradley,

Thank you very much for the opportunity to resubmit our manuscript. We are pleased that our manuscript is potentially acceptable for publication in BMC Emergency Medicine. Enclosed please find the point-to-point reply to the Editor’s and the reviewers’ comments and suggestions.

Thank you very much once again for your efforts improving our manuscript.

Yours sincerely,

Gaurav Nepal
On behalf of the authors

Technical Comments:

Remove duplicate sub-section in the declaration

Answer: We removed the duplicate sub-section from declaration section
Reviewer reports:

Reviewer 1:
The discussion and conclusion sections had improved compared with last version. Authors should correct minor language correction (particularly article, verb, and noun) through this manuscript, such as line 23 on page 5, it should be “Though” not "though", or the punctuation should be modified in line 22 on page 5.
Answer: Thank you for your valuable comments. We corrected the language errors.

Reviewer 3:
I have some minor comments and one regarding the term rapid ascent. The English is good but is in need of some correction before publication. Please see the uploaded attachment.
Answer: Thank you for your valuable comments. We explained the term rapid ascent and emergent ascent in the manuscript. We had a look at your attachment and addressed your comments in our revised manuscript.